Author's response to reviews

Title: Reducing errors in health care: cost-effectiveness of multidisciplinary team training in obstetric emergencies (TOSTI study); a randomised controlled trial

Authors:

Joost van de Ven (joost.vandeven@mmc.nl)
Saskia Houterman (s.houterman@mmc.nl)
Rob AJQ Steinweg (r.steinweg@mmc.nl)
Albert JJA Scherpbier (a.scherpbier@oifdg.unimaas.nl)
Willy Wijers (vk.onderwijs@mmc.nl)
Ben Willem J Mol (b.w.mol@amc.uva.nl)
S Guid Oei (g.oei@mmc.nl)
the TOSTI-trial group (b.w.mol@amc.uva.nl)

Version: 8 Date: 23 September 2010

Author's response to reviews: see over
Veldhoven, September 13th, 2010

Editor-in-Chief BioMed Central; Pregnancy and Childbirth

Subject: Cover letter study protocol “Reducing errors in health care: cost-effectiveness of multidisciplinary team training in obstetric emergencies (TOSTI study); a randomised controlled trial”.

Submitted earlier first version (July, 2009), second version (July 2010) and third version (August 2010): manuscript ID (1608797151287673)

Dear editor,

Herewith we submit our manuscript with the formatting changes as asked.

Following the formatting changes that are made:
- We removed all track changes in our manuscript
- We changed the title of our manuscript grammatically
- We removed the names of the authors from the title page
- We checked the typography of our manuscript

The manuscript can be categorized as a study protocol. The material has not been previously published. There are no personal conflicts of interest of any of the authors. Ethical approval is not required for this type of study in The Netherlands. The study is funded by ZonMw, the Dutch Organization for Health, Research and Development, project number 170992303.

We look forward to hear from you.

Yours sincerely,

Joost van de Ven, MD, corresponding author

Máxima Medical Centre Eindhoven-Veldhoven, the Netherlands
Dept of Obstetrics and Gynaecology
De Run 4600
P.O. Box 7777
5500 MB Veldhoven
The Netherlands
T: +31 (0)40 8888385
Tel: +31 641420951
E-mail address: j.vandeven1983@gmail.com / joost.vandeven@mmc.nl